Author’s response to reviews

Title: The importance of social environment in preventing smoking: An analysis of the Dead Cool intervention

Authors:

Jennifer Badham (research@criticalconnections.com.au; j.badham@qub.ac.uk)
Helen McAneney (h.mcaneney@qub.ac.uk)
Laura Dunne (l.dunne@qub.ac.uk)
Frank Kee (F.Kee@qub.ac.uk)
Allen Thurston (a.thurston@qub.ac.uk)
Ruth Hunter (ruth.hunter@qub.ac.uk)

Version: 5 Date: 10 Aug 2019

Author’s response to reviews:

I have clarified the ethics statement concerning who provided consent. The relevant sentence now reads:

Participation in the study is by voluntary informed written consent (school principals, teachers, opt-out for parents followed by opt-in for students), obtained by the evaluation researchers prior to all stages of data collection.